\begin{table}%t1
%\scriptsize{
%\begin{center}
\par
\caption{Pre-main sequence stars observed.}
\label{Stars}
\par
\begin{tabular}{lcccccrcc}
\par
\hline
\hline\noalign{\smallskip}
     Star  &   Location &       Ref. &      TTS$^{a}$ &  TDTnum$^{b}$ &  RA (2000) & Dec (2000) &        HBC &       IRAS \\\noalign{\smallskip}
% &  &  &  & \tiny{h~  m~ s~ } & \tiny{o~ '~ "~ } &  &  \\
\hline
     T Cha & Dcld 300.2-16.9 &          1 &          w &   11700504 & 11 57 13.53 & --79 21 31.5 &        591 & IRAS 11547-7904  \\
     Sz 68 &      Lup 1 &               2 &          w &   11801413 & 15 45 12.87 & --34 17 30.6 &        248 &  IRAS 15420-3408   \\
    RU Lup &      Lup 2 &               1 &          c &   11801914 & 15 56 42.31 & --37 49 15.5 &        251 & IRAS 15534-3740 \\
    EX Lup &      Lup 3 &               3 &          c &   11801715 & 16 03 05.48 & --40 18 25.9 &        253 &  IRAS 15597-4010 \\
    Sz 124 &      Lup 3 &               4 &          w &   11801816 & 16 11 53.35 & --39 02 15.7 &        631 &          -- \\
     ROX 2 & $\rho$ Oph / B42 &       4/5 &          w &   12101218 & 16 25 25.00 & --23 55 30.0 &        635 &          -- \\
   DoAr 24 & $\rho$ Oph / B42 &       4/5 &          c &   12101320 & 16 26 17.20 & --24 20 23.0 &        638 &          -- \\
  DoAr 24E & $\rho$ Oph / B42 &         5 &          w &   12001722 & 16 26 23.40 & --24 21 02.0 &        639 &          -- \\
     SR 12 & $\rho$ Oph / B42 &         5 &          w &   12101023 & 16 27 19.50 & --24 41 40.1 &        263 &          -- \\
    ROX 31 & $\rho$ Oph / B42 &         5 &          w &   12002025 & 16 27 52.00 & --24 40 48.0 &        642 &          -- \\
  Wa Oph/4 & L162=LDN162 &              6 &          w &   11400526 & 16 48 18.00 & --14 11 15.0 &        652 &          -- \\
  Wa Oph/6 & L162 / Oph &               6 &          c &   11400627 & 16 48 45.69 & --14 16 34.8 &        653 & IRAS 16459-1411 \\
    IX Oph &        B59 &               1 &          w &   12101729 & 17 09 48.16 & --27 16 59.3 &        272 &          -- \\
 V1725 Oph &        B62 &               4 &          w &   11801330 & 17 16 13.85 & --20 57 45.7 &        661 &          -- \\
CoKu Ser G1 & L572 / Ser MC  \Le&         4/7 &          w &   11401333 & 18 28 15.19 & --00 02 41.6 &        667 &          -- \\
CoKu Ser G7 &       L572 &              4 &          w &   11401535 & 18 29 56.56 & +01 14 49.1 &        672 & IRAS 18274+0112 \\
  V536 Aql &          -- &               -- &          c &   13000939 & 19 38 57.00 & +10 30 18.0 &        294 & IRAS 19365+1023 \\
   MacC H3 &    Cep OB4 &               4 &          c &   13100944 & 23 54 26.80 & +66 54 21.0 &        319 &          -- \\
\hline
\end{tabular}
\par
%\end{center}
\par
Notes to table: $^a$ See Table\ \ref{results}\ for corresponding 
{\it EW}(\Ha) and references. Classification w and c refer to \wt\ and \ct\ respectively.
$^b$~TDTnum is the assigned ISO observation number.
\newline
\textit{References:} (1)~\cite{Bertout99}, (2)~\cite{Geoffray01}, (3)~\cite{Appenzeller83}, (4)~Herbig-Bell Catalogue (HBC),
(5)~\cite{Bouvier92},
(6)~\cite{Walter86}, (7)~\cite{Jensen96}.
\end{table}